Author’s response to reviews

Title: Associations Between Intimate Partner Violence Profiles and Mental Health Among Low-income, Urban Pregnant Adolescents

Authors:

Jordan Thomas (thomasjl@ucla.edu)
Jessica Lewis (jessica.lewis@yale.edu)
Isabel Martinez (isabel.martinez@yale.edu)
Shayna Cunningham (shayna.cunningham@yale.edu)
Moiuri Siddique (ms3898@georgetown.edu)
Jonathan Tobin (jntobin@cdnetwork.org)
Jeannette Ickovics (jeannette.ickovics@yale.edu)

Version: 1 Date: 02 Jan 2018

Author’s response to reviews:

Dear Ms. Coore and Editorial Staff:

Thank you for your message about the status of our manuscript, "Associations Between Intimate Partner Violence Profiles and Mental Health Among Pregnant Adolescents" (PRCH-D-17-00609). We have completed the requested corrections, as outlined with our responses below:

1. Thank you for providing the Ethical Approval and Consent to Participate section in your manuscript. Please clarify whether written parental consent was obtained from participants under the age of 16 in this section.

We have added language to explain the protocol around parental consent in this section.

2. Please be sure to include each authors email address on the title page of your manuscript.

We have added author email addresses to the title page.

3. Please include a full list of abbreviations in the abbreviations section of your manuscript.

We have completed the full list of abbreviations in this section of the manuscript.
Thank you very much for your continued engagement. We look forward to hearing from reviewers about our manuscript.

Many thanks,

Jordan Thomas